Author’s response to reviews

Title: A family-based intervention targeting parents of preschool children with overweight and obesity: Conceptual framework and study design of LOOPS-Lund Overweight and Obesity Preschool Study.

Authors:

Jenny Önnerfält (jenny.onnerfalt@med.lu.se)
Lena-Karin Erlandsson (Lena-Karin.Erlandsson@med.lu.se)
Kristina Orban (kristina.orban@med.lu.se)
Malin Broberg (malin.broberg@psy.gu.se)
Christina Helgason (Christina.Helgason@psychology.lu.se)
Kristina Thorngren-Jerneck (Kristina.Thorngren-Jerneck@med.lu.se)

Version: 2 Date: 10 September 2012

Author’s response to reviews: see over